Author's response to reviews

Title: A randomized, double-blind, placebo-controlled trial of calcium acetate on serum phosphorus concentrations in patients with advanced non-dialysis-dependent chronic kidney disease

Authors:

WAJEH QUNIBI (QUNIBI@UTHSCSA.EDU)
WOLFGANG C WINKELMAYER (winkelmayer@stanford.edu)
RICHARD SOLOMAN (Richard.Solomon@vtmednet.org)
MOUSTAFA MOUSTAFA (mamoustafa@hotmail.com)
PAUL KESSLER (pkessler@nabi.com)
CHIANG-HONG HO (ChiangHong.Ho@fmc-na.com)
JONATHAN GREENBERG (jonathan.greenberg@fmc-na.com)
JOSE A DIAZ-BUXO (Jose.Diaz-Buxo@fmc-na.com)

Version: 3 Date: 27 December 2010

Author's response to reviews:

December 27, 2010

Dr. Melissa Norton - Editor in Chief
BMC Nephrology

Dear Dr. Norton,

We respectfully resubmit our manuscript entitled “A randomized, double-blind, placebo-controlled trial of calcium acetate on serum phosphorus concentrations in patients with advanced non-dialysis-dependent chronic kidney disease” by Wajeh Qunibi, Wolfgang C. Winkelmayer, Richard Solomon, Moustafa Moustafa, Paul Kessler, Chiang-Hong Ho, Jonathan Greenberg, and Jose A, Diaz-Buxo for your re-consideration for publication in BMC Nephrology.

The manuscript has been revised on the lines suggested by the editors and the reviewers. The changes in the revised version of the manuscript are “Tracked”. Additionally, we include an itemized list of responses for each of the Reviewer’s comments. We hope that our responses will satisfy the questions raised by the reviewers of BMC Nephrology.

Thank you for your consideration of our manuscript and I look forward to hearing from you soon.

Respectfully Yours,
Wajeh Qunibi, M.D.
Professor of Medicine
Medical Director of Dialysis
University of Texas Health Science Center
University hospital-Adult Renal Dialysis
4502 Medical Drive
San Antonio, Texas 78229
Phone: 210-422-2692
Pager: 210-553-3330
Fax: 210-358-4710
E-mail: qunibi@uthscsa.edu